**Author's response to reviews**

**Title:** Clinical and radiological outcome following treatment of displaced lateral clavicle fractures using a locking compression plate with lateral extension: a prospective study.

**Authors:**
- Marc Beirer (marc.beirer@gmx.de)
- Sebastian Siebenlist (sebastian.siebenlist@mri.tum.de)
- Moritz Crönlein (moritz.croenlein@mri.tum.de)
- Lukas Postl (lukas.postl@mri.tum.de)
- Stefan Huber-Wagner (stefan.huber-wagner@mri.tum.de)
- Peter Biberthaler (peter.biberthaler@mri.tum.de)
- Chlodwig Kirchhoff (chlodwig.kirchhoff@mri.tum.de)

**Version:** 3  
**Date:** 2 November 2014

**Author's response to reviews:** see over
Dear Dr. Catia Cornacchia,

Thank you very much for the kind consideration of our manuscript (Manuscript ID 1065071099143675) Clinical and radiological outcome following treatment of displaced lateral clavicle fractures using a locking compression plate with lateral extension: a prospective study.

and the valuable comments. We thoroughly revised the manuscript according to your suggestions. In the following we will explain the revisions that have been made. We have enclosed a point-by-point response. We are referring to the lines of the revised manuscript in this letter to the editor. Hence, it is a privilege to submit a revised version hereby.

Please clarify and state in the Methods section whether consent from patients for their data to be published in Table 1 was obtained.
We totally agree and therefore we added the missing information in the Methods section (lines 87-88):
All patients gave their written consent to publish their data in table 1.

Acknowledgements: We strongly encourage you to include an Acknowledgements section between the Authors contributions section and Reference list.
We added an acknowledgements section (lines 267-270) and stated that there was no source of funding for each person involved in the study (authors and acknowledgements).
We would like to explicitly thank Fritz Seidl, MA Interpreting and Translating and expert for statistics, for his excellent language copyediting and for his statistical analysis. The authors declare that they have no competing interests and that there is no source of funding.

We hope that our revised manuscript is now acceptable for publication in BMC Musculoskeletal Disorders.

Respectfully,
Dr. Marc Beirer, MD Dr. Chlodwig Kirchhoff, MD